Author’s response to reviews

Title: The Establishment and Utility of Sweha-Reg: A Swedish Population-based Registry to Understand Hereditary Angioedema

Authors:

Lotus Mallbris (lotus.mallbris@ki.se)
Patrik Nordenfelt (patrik.nordenfelt@telia.com)
Janne Bjorkander (janne.bjorkander@medfak.gu.se)
Anders Lindfors (Anders.Lindfors@ki.se)
Sonja Werner (Sonja.Werner@med.lu.se)
Carl-Fredrik Wahlgren (Carl-Fredrik.Wahlgren@ki.se)

Version: 3 Date: 8 October 2007

Author’s response to reviews: see over
Dear Editor,

Thank you for considering our manuscript, “The Establishment and Utility of Sweha-Reg: A Swedish Population-based Registry to Understand Hereditary Angioedema”, with ID 1500704175162376, for publication in BMC Dermatology in the format of a Correspondence type 3.

Please find below a point-by-point description of the changes made in the revised version of the manuscript that is submitted.

Page 2, Abstract

The previous text body of the abstract is changed to a structured text with titles of Background, Methods and Conclusions. In the last paragraph the sentence “The present manuscript seeks to raise awareness of the existence of Sweha-Reg” is extended and “to stimulate the international collaboration of registries. A synthesis of data from similar registries across several countries is required to approach an inclusive course understanding of HAE” is added.

Page 3, Introduction, second paragraph

The sentence “Although recent reports proposed increased activity of coagulation factor XII (Hageman factor) as a possible cause of HAE type III” and the relevant references are added.

Page 12, Authors' contributions

A detailed of authors' contributions is provided.

Thank you for considering our manuscript. I look forward to hear your final decision in the near future.

Sincerely

Lotus Mallbris